Author’s response to reviews

Title: Takotsubo cardiomyopathy in a patient with ileus: A case report

Authors:

Chen-Yu Guo (conradg0712@gmail.com)

Nan-Sung Chou (nansongchou23@gmail.com)

Version: 2 Date: 07 Oct 2017

Author’s response to reviews:

Comment 1: Please remove the writing from the figures, such as the hospital name and the date and time the images were taken.

Response: As requested, we removed the hospital name and the date and time the images were taken from the figures.

Comment 2: Please amend the Discussion heading to "Discussion and Conclusions", and remove the "Conclusions" heading, as standard for Case reports in the BMC Series.

Response: As requested, we amend the Discussion heading to "Discussion and Conclusions", and remove the "Conclusions" heading in the revised manuscript.

Comment 3: Please remove the CARE checklist, as it is no longer needed, and include a statement in the Case presentation section stating that the CARE guidelines were followed in this report.

Response: As requested, we removed the CARE checklist and include a statement in the Case presentation section stating that the CARE guidelines were followed in this report in the revised manuscript.

Comment 4: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Response: As requested, we uploaded the revised manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant figure files are also clean versions and uploaded as separate files.